Author's response to reviews

Title: Posttraumatic Stress Disorder: An exploratory study examining rates of trauma and PTSD and its effect on outcomes in Community Mental Health

Authors:

Irene M Howgego (irene.howgego@act.gov.au)
Cathy Owen (cathy.owen@anu.edu.au)
Lenore Meldrum (lenore.meldrum@bigpond.com.au)
Peter M Yellowlees (pmyellowlees@ucdavis.edu.us)
Frances Dark (Frances_Dark@health.qld.gov.au)
Ruth Parslow (ruth.parslow@anu.edu.au)

Version: 2 Date: 6 December 2004

Author's response to reviews:

This is a revised version with tables added within this document rather than the figures document